RF02 

TECH: Cowan
4/21/08

1747 Block out

1752 Take off

1819 Troubleshooting MEDUSA Flow. Found that there was nothing wrong with our cables.

1829 THIMIL1 looks very small less than .1 V.

1850 Turning off THIMIL1 to see if it makes a difference in the reading.

1850 Seeing a difference, display must be messed up, should be in C not V.

1855 Power Readings:

Left A Bus

400 A Phase 11.5 – 16 A

400 B Phase 10.5 A 

400 C Phase 12.3 A

400  Phase A 115 Volts

400  Phase B 116 Volts

400  Phase C 115 Volts

400 AI

All 0 with 115V

60Hz  117 V  12 A

28Volts AI  29.1 V 9 A

Right Side B Bus

400 A Phase 7.4 A

400 B Phase 12.6 A

400 C Phase 9.8 A

400  Phase A 115 Volts

400  Phase B 113 Volts

400  Phase C 115 Volts

400 AI

A 0

B 3.9A

C 0

All 115V

60Hz 115 V 18 A

1913 Does not appear that the slow NOAA O3 is displaying correctly.

1933 Big spike in PLWC. DSM temp still well above 0C so that is not the issue.


Good until about 1853 then a shift down to 0W. Another spike at 1812. 
1939 Lost all ground chat clients.

2240 Large PLWC spike, spikes are not tied to the element since it appears broken.

2335 PLWC very large spike for a long time, shorter spike at 2243 also.


Spike was 2 minutes or more in length.

0019 Another large PLWC spike staying at 70W.

0021 Cycled power on LWO DSM hoping to clear PLWC problem.

0024 DSM back up, probes rebooted.

0025 When DSM was reset about 2 W on PLWC,, looks like the 0W was a DSM problem and not an element problem.

0107 Seeing a streaker or stuck bit on the 2DC. Resetting the DSM and the probes did not clear the problem.

0153 Landing

Post Flight:

On data system shutdown the system did not shutdown correctly.  It stopped at UNLOADING IPTABLES MODULE.

